Author's response to reviews

Title: Severe anemia due to pharyngeal leech infestation; A case report from Ethiopia

Authors:

Nebiyu Shitaye (handworklove@gmail.com)
segenet shibabaw (segenetlove@gmail.com)

Version: 1 Date: 05 Jul 2017

Author’s response to reviews:

To editor; I verify the manuscript again to minimize similarity score based on your recommendation..

To reviewer1: thank you for your constructive comments.

1. Language editing and spelling error; some one fluent in English commented and improved. One exception to this is; I persist to write gm/dl against to your comment mg/dl because that is how hemoglobin level is correctly expressed/measured.

2. Based on your recommendation other possible differential diagnosis that gives the clinical picture of the patient is discussed in one paragraph

For reviewer2: thank you for your nice comments

1. Best technique to remove leeches upper aerodigestive tract is discussed in the last paragraph of the discussion part

2. The reference is corrected according to the journal referencing style

For reviewer3: I truly appreciate the support and networking contact you have given to me so that we can improve the English language to some extent